Author’s response to reviews

Title: A retrospective analysis of survival and prognostic factors of male breast cancer from a single center

Authors:

Amr A Soliman (amr_soliman@alexmed.edu.eg)
Adel T Denewer (adeldenewer@mans.edu.eg)
Wael El-Sada (waelsadda@gmail.com)
Ali H Abdel-Aty (alihosniali@mans.edu.eg)
Basel Refky (dr.basel@hotmail.com)

Version: 7 Date: 24 March 2014

Author’s response to reviews: see over

Subject: Re-Submission after language editing by native speakers

“A retrospective analysis of survival and prognostic factors of male breast cancer from a single center”.

Dear Ms. Patricia Ratin,

I express my gratitude on behalf of the authors for considering and accepting our manuscript for publication in your journal, and hereby would like to re-submit a modified version as suggested in your response.

The changes you indicated, as regards funding, are integrated in the acknowledgments section of the manuscript, and the submitted version is as provided after language editing by native speakers.

Thank you once more for considering and accepting our work.

On behalf of the authors

With best regards

Dr. Amr A Soliman

Contact details of the corresponding Author:
Amr Soliman
Address: El-Shatby Maternity University Hospital
Port Said street, Shatby.
21526, Alexandria, Egypt.
Tel. 002 03 546 5034
Fax. 002 03 486 3981
Email: amr_soliman@alexmed.edu.eg; solimansaamr@yahoo.com.